Author's response to reviews

Title: Acute effect of meal glycemic index and glycemic load on blood glucose and insulin responses in humans

Authors:

Jose J Galgani (jgalgani@inta.cl)
Carolina C Aguirre (caguirre@inta.cl)
Erik E Diaz (ediaz@inta.cl)

Version: 4 Date: 20 June 2006

Author's response to reviews:

Please, add the following corrections to the just sent revision.
6. We arranged the table 2. We deleted the "a" letter from the followings rows. tAUC FFA (2h), tAUC glucagon (2 and 5 h), given no differences were detected among the meals.